Author's response to reviews

Title: Socioeconomic status, urbanization and risk behaviors in Mexican youth: an analysis of three cross-sectional surveys

Authors:

Juan Pablo Gutiérrez (jpgutier@insp.mx)
Erika E Atienzo (eeatienzo@insp.mx)

Version: 10 Date: 11 November 2011

Author's response to reviews: see over
Dear editorial board:

We are submitting a revised version of the manuscript 2712974524208514: "Socioeconomic status, urbanicity and risk behaviors in Mexican youth: an analysis of three cross-sectional surveys". In this new version, the two tables have been formatted as suggested by The BMC Series Editorial Production. They were created using Excel.

We look forward to hearing from you soon,

Erika E. Atienzo, MS (Corresponding author)
e-mail: eeatienzo@insp.mx

Mailing address: Instituto Nacional de Salud Pública. Dirección de Salud Reproductiva.
Av. Universidad No. 655 Col. Santa María Ahuacatitlán.
CP. 62100.
Cuernavaca Morelos, México.
Phone: +52 (777) 329 30 00 ext. 3113